Author’s response to reviews

Title: Seasonal and Climatic Variation in the Incidence of Adult Acute Appendicitis: A Seven Year Longitudinal Analysis

Authors:

Thomas York (thomas.york@nhs.net; tom.york@live.com)

Version: 2 Date: 22 Mar 2020

Author’s response to reviews:

In addition to the changes described below, the minor changes detailed in the email dated the 20/03/2020 have been completed in full.

- All abbreviations are now given in full in the first instance and subsequently included in brackets
- ‘Consent for publication’ section has been changed to state not applicable
- ‘Introduction’ section title has been changed to background in both the abstract and body of the manuscript
- Manuscript uploaded in a clean form